EO-1 Weekly Status Week of July 20 – July 26, 2007

Day of Year 201 - 207
Mission Day 2431 - 2437
Earth Observing-One (EO-1) General:

There were 126 Data Collection Events (DCEs) scheduled this week.  
Instruments underwent decontamination cycling as follows:

The Hyperion hyperspectral imager (HSI) performed a deicing from 07/20/2007 at 01:25z (07-201) to 07/20/2007 at 16:15z (07-201).  

The Advanced Land Imager (ALI) performed an outgassing from 07/20/2007 at 01:35z (07-201) to 07/20/2007 at 16:25z (07-201).  

EO-1 Spacecraft Subsystems:

All spacecraft subsystems functioned nominally with the exception of the Command & Data Handling (C&DH) telemetry playback anomaly reported previously.  
Instruments
All instruments operated nominally this week.  
Technology Activities:
In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

Sensor Web & Virtual Observatory Demonstrations

Another teleconference was held between the Wildfire Research and Applications Program (WRAP) and the Advanced Information System Technology (AIST) Sensor Web collaborators on Thursday, 26 July.  The following key activities being performed by the collaborators were discussed:
· There are being tested several workflows that string together sensor web services to accomplish particular science goals.

· Workflows being implemented are as follows:

· Tasking of EO-1
· Tasking of ASTER instrument on Terra 

· Tasking of Ames UAS

· Delivery of user customized fire related data products.

· Work has begun on tying together these workflows so that manual action to start any one is not required.

· GMU delivered a BPEL-based workflow that retrieves Hyperion Level 1G data from the GeoBliki SOS , runs the JPL thermal classifier WPS, ingests that result into a Web Coverage Service-Transaction (WCS-T) entity, and then drapes it over a map supplied by a Web Map Service (WMS).  This workflow will be tested next week and integrated with the EO-1 tasking workflow.
For the ongoing GEOSS activity, information has been shared about the new sulfur detection algorithm that is now operational onboard EO-1 and about our science collection of ground truth data in Greenland and Ellesmere Island in Canada with the GEOSS architectural implementation Pilot working group on biodiversity and environmental monitoring for the International Polar Year.
Dan Mandl presented a paper entitled “Experiments with User-Centric GEOSS Architectures” at IGARSS 2007 in Barcelona, Spain on July 23, 2007.  The session was entitled “GEOSS Challenges and Opportunities” and was attended by many people from the international GEOSS community.

Autonomous Sciencecraft Experiment (ASE)

ASE controlled EO-1 all week.  The new ASE (R4) code has been operational since 2007/199.  Certain tests are being conducted to verify the additions to the new code.  The image "Borup Fiord Pass [JPL/W6]" with the scene name "EO10570012007205110KF_94610614” has been captured on orbit 35511 at 2007-205/20:04:12z to validate the new sulfur classifier and confirm the results of the science algorithm executed on board.  

Ground System

Hardware Status

	Hardware
	Status
	Comments

	Primary R/T System
	
	Green
	

	Backup R/T System
	
	Green
	FOT has upgraded ASIST to 9.7.e on the backup R/T system.  Testing is currently underway.  Telemetry and commanding will be tested utilizing the flight software lab and Flatsat.  The backup system is currently setup as an associate ASIST station and will continue to shadow the pass schedule.

	bMOC at DataLynx
	
	Green
	

	Playback System
	
	Green
	

	Mission Planning
	
	Yellow
	EO-1’s DBA is scheduled to setup the mission planning Oracle database on Axle after the security audit.  Mission planners will continue to perform all tasks using Speedwheel and Bullseye.

	Flight Dynamics
	
	Green
	All licensing issues have been resolved.  All required software, with the exception of EO-1’s custom code, “auto-products”, has been installed on both new systems.  Developer Bob Wiegand is tasked with the upgrade of “auto-products”.  FOT will continue to verify the compatibility of the new setup to the old systems and proceed with parallel operations.  

	DTAS Server
	
	Yellow
	As the result of a preliminary security scan, a part of the DTAS trending system code has been flagged as a high risk element.  The server is currently on the network and will be populated with data manually by the FOT using external data storage units.  FOT and the system administrators are discussing the options of how to keep the server on the network.  

	ANS
	
	Green
	

	Flight Software Lab
	
	Green
	

	Admin System
	
	Green
	


Operations
NASA Security Audit

As preparation for the upcoming security audit, EO-1s system administrator has been implementing certain changes to the systems to ensure that EO-1 FOT follows the NASA guidelines for network and system security.  Listed below is a summary of these modifications.
1. Installed anti-virus and spy ware software on Waldorf2 and Statler2.

2. All the Banners (FTP, SFTP, SSH, RLOGIN, TELNET, GDM) on all the systems were updated. The previous warning banners were out dated.

3. Waldorf, the Windows NT system was updated with AVG anti-virus software. The McAfee software provided by CNE does not support Windows NT.  The AIX system (Shelf) is not scheduled to be audited but it has also been updated.

4. Sling2 (Linux) has been equipped with CLAMAV, the suggested anti-virus software.  This system will be the test bed system for the Linux anti-virus software until testing verifies that the anti-virus software does not break/interrupt mission operations.

EO-1 is scheduled to have 15 stations scanned plus the trending system DTAS which is currently off the network.  Below is a listing of the 15 nodes.
	DESCRIPTION
	SPECIFICATION
	QUANTITY
	LOCATION
	CODE
	PROJECT 

	EngSci. Workstation
	Dec Alpha Workstation
	2
	Build 14/rm N218
	428
	 E0-1MOC

	EngSci. Workstation
	Fedora Core Linux
	1
	Build 14/rm N218
	428
	 E0-1MOC

	EngSci. Workstation
	HP UX 10.20 Unix
	2
	Build 14/rm N218
	428
	 E0-1MOC

	EngSci. Workstation
	Windows NT Workstation
	2
	Build 14/rm N218
	428
	 E0-1MOC

	EngSci. Workstation
	Windows XP
	2
	Build 14/rm N218
	428
	 E0-1MOC

	EngSci. Workstation
	Red Hat 3 Enterprise
	6
	Build 14/rm N218
	428
	 E0-1MOC


New Discrepancy Reports (DR)

There are no new requests this week.
New Enhancement Requests (ER)
There are no new requests this week  

New Operations Database Change Requests (DBCR)

There are no new requests this week.
Ground and Space Network

EOS Data and Operations System (EDOS), GSFC, MD, USA

There are no major problems to report.
McMurdo Ground Station (MGS), McMurdo, Antarctica

There are no major problems to report.
Wallops Ground Station (WGS), Wallops Island, VA, USA

There are no major problems to report.
DataLynx Ground Station (PF1), Poker Flat, AK, USA

There are no major problems to report.
Alaska Ground Station (AGS), Poker Flat, AK, USA 

There are no major problems to report.
Svalbard Ground Station (SGS), Longyearbyen, Norway
There are no major problems to report.
Tracking and Data Relay Satellite System (TDRSS), White Sands, NM, USA

There are no major problems to report.
USGS Multi-Satellite Ground Station (LGS), Sioux Falls, SD, USA
LGS is now being scheduled only for contingencies.
Tasmanian Earth Resources Satellite Station, a.k.a. Hobart (HGS), Hobart, Australia

Australian Centre for Remote Sensing reports that TERSS has been having technical issues.  The antenna will be down until further notice.  Original estimates listed a return to operations in early July.  The FOT requested an update on the status of the HGS repairs.  Engineers arrived at Hobart with new components on July 17, 2007.
Cordoba Ground Station (CGS), Cordoba, Argentina

No passes were scheduled this week. 

Upcoming Events:

· Lunar calibration to be conducted on July 30, 2007

· Security audit tentatively scheduled for July 30, 2007

· AC Solar calibration to be conducted on August 1, 2007

· HSI Solar calibration to be conducted on August 1, 2007

Imagery Status:

Scenes and Engineering Cals planned for week of July 20 – July 26, 2007                          126
Total scenes and engineering calibrations planned for entire mission – approximately 35,275
Total Scenes:  ALI scenes in the Level 0 archive            31,942 (as of July 26, 2007)

                        Hyperion scenes in the Level 0 archive    31,689
Publications and Presentations Status (as of 07-05-07):

404 publications 

284 external presentations

51 articles and press releases
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